Author's response to reviews

Title: One size does not fit all: A qualitative content analysis of the importance of existing quality improvement capacity in the implementation of Releasing Time to Care: The Productive Ward™ in Saskatchewan, Canada.

Authors:

Jessica Hamilton (jhamilton@hqc.sk.ca)
Tanya Verrall (tverrall@hqc.sk.ca)
Jill Maben (jill.maben@kcl.ac.uk)
Peter Griffiths (peter.griffiths@soton.ac.uk)
Kyla Avis (kavis@hqc.sk.ca)
G. Ross Baker (ross.baker@utoronto.ca)
Gary Teare (gteare@hqc.sk.ca)

Version: 2
Date: 12 March 2014

Author's response to reviews: see over
March 12, 2014

Re: 1039583630122019 - One size does not fit all: A qualitative content analysis of the importance of existing quality improvement capacity in the implementation of Releasing Time to Care: The Productive Ward™ in Saskatchewan, Canada.

Editor-in-Chief
BMC Health Services Research

Thank you for the opportunity to submit a reformatted version of the research article entitled “One size does not fit all: A qualitative content analysis of the importance of existing quality improvement capacity in the implementation of Releasing Time to Care™ in Saskatchewan, Canada” by Jessica Hamilton, Tanya Verrall, Jill Maben, Peter Griffiths, Kyla Avis, Ross Baker and Gary Teare for consideration for publication in BMC Health Services Research. Releasing Time to Care: The Productive Ward™ was implemented in Saskatchewan, Canada between 2008 and 2012 and this manuscript describes a component of a research evaluation that occurred parallel to its implementation.

Your comment asked us to include a statement about whether informed consent was received from the interview participants. Please note that we have addressed this point on lines 169-172 of the manuscript and have indicated that written consent was obtained from all participants.

Thank you for your consideration.

Sincerely,

Jessica Hamilton
Researcher
Health Quality Council, Saskatchewan